Author’s response to reviews

Title: Self-medication with over the counter drugs, prevalence of risky practice and its associated factors in pharmacy outlets of Asmara, Eritrea

Authors:

Sirak Tesfamariam (sirakino2015@gmail.com)
Indermeet Anand (drisanand@gmail.com)
Ghide Kaleab (ghidekaltes@gmail.com)
Samson Berhane (fevengezaea@gmail.com)
Biruck Woldai (brookpharm@gmail.com)
Eyasu Habte (yohoshua79@yahoo.com)
Mulugeta Russom (satiswt@gmail.com)

Version: 5 Date: 18 Jan 2019

Author’s response to reviews:

Figure 1 is cited at its appropriate place. Moreover, "Additional files" section is added and the necessary information is provided as requested.